Author’s response to reviews

Title: Genotype-phenotype correlation analysis of MYO15A variants in autosomal recessive non-syndromic hearing loss

Authors:

Jing Zhang (Baggio0812@163.com)
Jing Guan (ggy3u@126.com)
Hongyang Wang (whyx301@foxmail.com)
Linwei Yin (yinlinwei@163.com)
Dayong Wang (Wangdy301@163.com)
Lidong Zhao (plagh@126.com)
Huifang Zhou (zyyzhf@163.com)
Qiuju Wang (wqcr301@vip.sina.com)

Version: 3 Date: 18 Mar 2019

Author’s response to reviews:

Dear editor:

Thank you very much for the editing and correction you have done on our paper, we have made the changes according to your comments.

Thank you very much for your attention and consideration.

Sincerely,

Qiuju on behalf of all of authors

Response to the editor:

1. It has come to our attention that within your Methods there is significant text overlap with other publications. Please seen the attached screen shot as an example of this overlap. While we understand that you may wish to express some of the same ideas contained in these publications,
please be aware that we cannot condone the use of text from previously published work. Please re-phrase this section in your own words to avoid direct overlap (please note, we cannot accept direct textual overlap with any previously published works even if the authors of the work are yourselves). Please be informed that we cannot proceed with handling your manuscript before this issue is resolved, and the sections of text in question have been reformulated.

Reply to the editor: We have re-phrased Methods section in our own words to avoid direct overlap of manuscript according your suggestion. (page 7 to 9)

2. The individual contributions of all authors to the manuscript should be specified in the Authors’ Contributions section. Currently HZ is not present.

Reply to the editor: Thanks for your suggestion. We have specified the individual contributions of all authors to the manuscript in the Authors’ Contributions section. (page 19 line 50)

3. Please provide the heading "Declarations" for the corresponding section.

Reply to the editor: Thanks for your suggestion. We have added the heading "Declarations" for the corresponding section. (page 18 line 59)

4. Thank you for providing your highlighted manuscript. However, at this stage it is not required and so we kindly ask that you remove it from your manuscript.

Reply to the editor: Thanks for your suggestion. We have deleted the highlighted manuscript.